\documentclass [12pt] {article}


\usepackage{graphicx, amsmath}
\DeclareGraphicsRule{.tif}{png}{.png}{`convert #1 `dirname #1`/`basename #1 .tif`.png}


\topmargin 0pt
\advance \topmargin by -\headheight
\advance \topmargin by -\headsep
\textheight 8.9in
\oddsidemargin 0pt
\evensidemargin \oddsidemargin
\marginparwidth 0.5in
\textwidth 6.5in

\begin{document}
\begin{center}

\Huge{Quiz 0}
\end{center}
\vskip .25in
\normalsize

This is a two-stage quiz.� You will be given this sheet of paper twice.  

The first (colored) copy of the quiz corresponds with the first stage.� Use your knowledge \& calculator to take this quiz--no notes, books, internet resources, faculty, tutors or colleagues. You have 15 minutes.

The second (white) copy of the quiz corresponds with the second stage. You are now welcome to use your books, notes, and students in the class to retake the same quiz. You have 15 minutes to complete the quiz and to build one solution to be turned in (with everyone's name on top!).� 

\vspace{.5in}

\noindent{Show \emph{all} your work. Reasonable supporting work must be shown for any partial credit. }

\begin{enumerate}
\item $[4]$ Write down something that the person/people taking this quiz are good at. \\
\vspace{1.25in}

\item $[4]$ Explain how the person/people got good at it! \\
\vspace{1.25in}

\item $[2]$ Let $a$ and $b$ be real, non-zero numbers. Add $\displaystyle{\frac{1}{a}+\frac{1}{b}}$.


\vspace{1.0in}

\end{enumerate}
\end{document}
